Author’s response to reviews

Title: LONG-TERM VISUAL OUTCOMES AND REHABILITATION IN USHER SYNDROME TYPE II AFTER RETINAL IMPLANT ARGUS II.

Authors:
Maria Iglesias (mariaiglesiasalvarez@gmail.com)
Maria Iglesias (mariaiglesiasalvarez@gmail.com)

Version: 6 Date: 06 Jul 2018

Author’s response to reviews:

RESPONSE TO THE COMMENTS

BOPH-D-17-00591R1

LONG-TERM VISUAL OUTCOMES AND REHABILITATION IN USHER SYNDROME TYPE II AFTER RETINAL IMPLANT ARGUS II.

Jeroni Nadal; Maria Iglesias

BMC Ophthalmology

Dear Amy Branch-Hollis

We would like to thank you for the last review. We believe that with your comments the quality of the article improves in order to be finally accepted for its publication.
Kind regards,

Maria Iglesias

Jeroni Nadal

Responses:

1. Thank you for clarifying that you have obtained consent for publication from the interpreter in the video. Please remove the consent form from the file inventory, as this is no longer needed.

Response:

Removed.

2. Please ensure that the text that is included in the video is translated into English.

Response:

We have included a final version without text. In Video 1 you can see the patient and his interpreter speaking in sign language. There is no need to translate the conversation since the aim of the video is to see the fast intervention in the conversation after the surgery.

3. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.

Response:

A clean final version of the manuscript is included.